Branley-Bell and Talbot *Journal of Eating Disorders* https://doi.org/10.1186/s40337-020-00319-y

(2020) 8:44

### Journal of Eating Disorders

#### RESEARCH ARTICLE

**Open Access** 

# Exploring the impact of the COVID-19 pandemic and UK lockdown on individuals with experience of eating disorders



Dawn Branley-Bell\* and Catherine V. Talbot

#### **Abstract**

**Background:** The coronavirus disease 2019 (COVID-19) pandemic may raise unique challenges for individuals with experience of eating disorders. Many factors have potential for detrimental impacts on psychological wellbeing and eating disorder recovery, including: Disruption to living situations, 'social distancing' restrictions, difficult access to healthcare, and societal changes to food behaviours and technology usage. To date, little is known on the impact of the pandemic on this population, particularly within the UK.

**Method:** A mixed-methods online survey was developed for the purpose of this study. Data was collected from 129 individuals currently experiencing, or in recovery from, an eating disorder during the early stages of the UK pandemic lockdown. Participants were aged between 16 and 65 years, with 121 participants identifying as female, 7 male and 1 participant preferring not to disclose their gender.

**Results:** Findings suggest that the pandemic is having a profound, negative impact upon individuals with experience of eating disorders. Eight key themes were generated: Disruption to living situation, increased social isolation and reduced access to usual support networks, changes to physical activity rates, reduced access to healthcare services, disruption to routine and perceived control, changes to relationship with food, increased exposure to triggering messages, and positive outcomes. The results suggest detrimental impacts on psychological wellbeing including decreased feelings of control, increased feelings of social isolation, increased rumination about disordered eating, and low feelings of social support.

**Conclusions:** Individuals with eating disorders are at significant risk of negative impacts of the pandemic. There is a vital need for interventions to support this population. Inequalities in healthcare provision were identified, emphasising a need for a more cohesive approach to remote treatment across UK healthcare services. Positive aspects of technology use were identified but the results suggest a need to address and/or limit the potential for negative impacts of public messages around food and exercise behaviours, and to co-design technologies with end-users to facilitate effective treatment.

**Keywords:** Pandemics, Coronavirus, COVID-19, Eating disorders, anorexia nervosa, Bulimia nervosa, Vulnerable populations, Healthcare

<sup>\*</sup> Correspondence: dawn.branley-bell@northumbria.ac.uk Department of Psychology, Northumbria University, Newcastle upon Tyne, England, UK



#### **Plain English summary**

This research uses an online survey to gather data from individuals in the UK with experience of eating disorders during the coronavirus disease 2019 (COVID-19) pandemic. The results suggest that disruptions to daily life as a result of lockdown and 'social distancing' may have a negative impact on individuals' wellbeing. Negative effects may be due to changes to individuals': regular routine, living situation, time spent with friends and family, access to treatment, engagement in physical activity, relationship with food and use of technology. The research discusses how these issues can be addressed via further developments within healthcare, research, governance and policy. This could benefit those experiencing eating disorders and also mental health issues more broadly.

#### Introduction

Individuals with, or in recovery from, eating disorders (EDs) are likely to be negatively impacted by the COVID-19 pandemic. ED symptoms may be exacerbated by media exposure, disruption to daily activities, social isolation, modified physical activity and sleep, and negative affect and fear of contagion [1]. The pandemic may also lower protective capacity and access to care, by restricting access to social support, treatment, and adaptive coping strategies [1]. Psychological factors linked to ED symptoms will likely be exacerbated by the pandemic and associated lockdown. For example, stress [2, 3], health anxieties and contamination fears [2, 4, 5], perceived lack of control and problems with emotional regulation [5, 6], reliance upon routine, structure [4, 5], and social isolation [3, 5].

Technology may play an important role at this pivotal time when physical socialisation is limited and virtual interaction comes into its own. Internet use has been linked to both negative and positive consequences for this population [7]. Thus, it is important to investigate the role that technology is playing at the current time. As a result of the pandemic, healthcare organisations also find themselves suddenly relying more heavily on virtual methods to deliver treatment and support. Research is already underway to identify how therapy for EDs can be translated to online delivery [8, 9]. Despite these efforts, healthcare organisations - and those receiving treatment - may face significant challenges with the move to remote healthcare [4]. For a population that can require regular in-person appointments [5], this reduction in service provision could be catastrophic.

Researchers from across the globe have started to investigate the impact of the pandemic upon individuals experiencing EDs. Early studies from Singapore and Barcelona highlight how increased anxiety is having a negative impact [2, 10]. In Australia, Phillipou et al. [11] found evidence of increased restricting, binge eating,

purging, and exercise among this population. Many reports focus upon countries with greater experience of infectious disease outbreaks (e.g., [4, 5, 12]). These countries are generally more prepared for dealing with the pandemic than others [5] due to previous experience of contact tracing and disease surveillance, provision of emergency healthcare resources needed for isolation, and co-ordinated public health messaging. There are other global studies in progress, e.g., in Australia [11], the United States [13] and the Netherlands [13]. However, research within the United Kingdom (UK) is limited. Fernández-Aranda et al. [10] conducted a small, qualitative UK study (N = 8) describing some of the challenges faced by those with EDs during COVID-19. However, no quantitative data was reported, and the sample was specific to those with Anorexia. To the best of our knowledge, there has been no empirical work conducted with people with EDs in the UK during the pandemic. The current research addresses this, with the aim of investigating the impact of the pandemic and UK lockdown on the lives of people with experience of EDs. In addition to raising awareness, the results could influence future health service provisions, guidance and policies.

#### Methods

#### Sample

One hundred fifty-three participants were recruited via social media (Twitter and Facebook) using opportunistic and snowball sampling. Inclusion criteria was: Over 16 years of age, current UK resident, experience of an ED (including those in recovery). Recruitment was limited to the UK to allow the researchers to accurately identify the stage of the pandemic and the lockdown, and to ensure that data was comparable in this regard. Twentyfour participants were excluded due to not proceeding past the first page of the survey or not meeting the inclusion criteria. The final sample consisted of 129 participants, aged between 16 and 65 years (M = 29.27yrs., SD = 8.99 yrs). One hundred twenty-one participants identified as female (93.8%), 7 male (5.4%) and 1 (0.8%) participant preferring not to disclose their gender. The majority (N = 80, 62%) of the sample described themselves as currently experiencing an ED (including those in a period of relapse), 8 (6.2%) described themselves as "in recovery <3 months", 8 (6.2%) "in recovery between 3-12 months", and 33 (25.6%) "in recovery >12 months".

#### Procedure and measures

Recruitment took place in early April 2020, 2 weeks after UK lockdown restrictions were first enforced. Participants completed an anonymous online survey containing questions around their experience of EDs and the pandemic. This included closed and open-ended questions about: the social impact of the lockdown; internet usage;

food and exercise behaviours; and the overall impact of the pandemic on their EDs. Established scales were used to measure: Mental wellbeing, perceived stress, social support, perceived control and rumination (see Supplementary Material 1).

#### Short Warwick-Edinburgh mental wellbeing scale

The 7-item Short Warwick-Edinburgh Mental Wellbeing Scale (SWEMWBS: [14, 15]) was used to measure emotional and functioning aspects of mental wellbeing. Scores range from 7 to 35, with higher scores indicative of greater mental wellbeing. Cronbach's alpha for the scale in the current study = .81.

#### Perceived stress scale

The Perceived Stress Scale (PSS: [16]) is one of the most widely applied measures of perceived stress, i.e., the degree to which an individual perceives situations within their life as stressful. The shortened 4-item version was used in the current study as it has been shown to maintain good validity and reliability [17]. The PSS-4 gives a score out of 16, with higher scores indicating greater perceived stress. Cronbach's alpha = .72.

#### **ENRICHD** social support instrument

The ENRIHD Social Support Instrument (ESSI: [22, 23]) is a 7-item measure of social support, i.e., the existence or availability of people on who an individual can rely. Scores range between 8 to 34, with higher scores indicating greater social support. Cronbach's alpha = .86.

#### Shapiro control inventory

To measure perceived control we used the Shapiro Control Inventory (SCI, [24]) general domain sense of control scale. This 16-item scale encompasses positive and negative sense of control. Scores range between 16 and 112 with higher scores indicating greater perceptions of control. Cronbach's alpha = .86.

#### Rumination response scale for ED

The Rumination Response Scale for Eating Disorders (RRS-ED, [25]) is a 9-item measure of preoccupation with ruminative themes of eating, weight and shape. It is the first self-report measure to specifically capture ED rumination. The measure gives a total score between 9 and 36, with higher scores indicative of greater rumination. Cronbach's alpha = .86.

#### Analytical approach

A mixed methods approach was applied. Descriptive statistics, and between group comparisons (current ED group vs. recovery group) were complemented with thematic analysis of the qualitative data (using NVivo12). The second author coded the dataset independently

using both an inductive and deductive approach, whereby some categories were developed prior to the analysis and others were generated from the data. Codes were collated to identify initial themes. To increase rigor, a team approach was taken with the first author providing critical feedback on themes and coding. Amendments were made where appropriate. The finalised themes were then generated and agreed upon by both authors.

#### Results

#### Quantitative measures

Mean scores for the five scales are provided for participants categorised into two groups: those currently experiencing an ED, and those who report being in recovery (Table 1).

Significant differences were found between the two groups for all measures except the rumination scale which just failed to reach significance (p = .052). The group differences are in the direction that would be expected with the recovery group showing higher mental wellbeing, lower perceived stress [2, 3], higher social support [3, 5] and higher perceived control [5, 6]. The borderline significant result was also in the expected direction with those in recovery scoring lower for ED rumination [26].

#### Overall impact of the pandemic

When asked about the overall impact of the pandemic on their ED symptoms, an overwhelming majority (86.7%) reported that their symptoms had worsened as a result of the pandemic, with over 30% reporting that their symptoms were *much* worse. In contrast, only 2 participants reported a slight improvement in their symptoms (Table 2).

We explore potential factors for this detrimental impact further in the following section.

#### Themes

The analysis generated eight themes relating specifically to the effects of the pandemic and associated lockdown:
(1) Disruption to living situation; (2) Increased social isolation and reduced access to usual support networks;

**Table 1** Participant scores for mental wellbeing, perceived stress, social support, perceived control and ED rumination

|                           | Current ED |         |       | In Recovery |         |       |
|---------------------------|------------|---------|-------|-------------|---------|-------|
|                           | N          | М       | SD    | N           | М       | SD    |
| Mental Wellbeing (WEMWBS) | 80         | 16.35*  | 3.24  | 49          | 17.66*  | 2.48  |
| Perceived Stress (PSS)    | 80         | 14.66** | 2.61  | 49          | 13.10** | 2.71  |
| Social Support (ESSI)     | 80         | 19.76*  | 5.71  | 49          | 21.94*  | 6.47  |
| Control (SCI-general)     | 77         | 54.52** | 13.15 | 48          | 62.71** | 12.74 |
| Rumination (RRS-ED)       | 76         | 22.67†  | 6.96  | 47          | 20.04†  | 7.59  |

Significant differences between the groups are indicated. \*\* p = .001 \* p < .05 †p = borderline (.052)

**Table 2** Participants' reported overall impact of the pandemic on their ED symptoms (N = 129)

|                      | N  | %    |
|----------------------|----|------|
| Symptoms much worse  | 38 | 30.9 |
| Moderately worse     | 48 | 37.2 |
| Slightly worse       | 24 | 18.6 |
| No change            | 11 | 8.5  |
| Slightly better      | 2  | 1.6  |
| Moderately better    | 0  | 0    |
| Symptoms much better | 0  | 0    |
| Missing              | 6  | 4.7  |

(3) Changes to physical activity rates; (4) Reduced access to healthcare services; (5) Disruption to routine and perceived control; (6) Increased exposure to triggering messages; (7) Changes to the individual's relationship with food; (8) Positive outcomes.

#### Disruption to living situation

Over a fifth of participants reported a change to their normal living situation due to the pandemic (Table 3). When asked whether they felt this change had affected their ED symptoms, 85.2% reported that the change had worsened their symptoms. Reasons included: Increased interpersonal stresses, hiding their ED from others, increased scrutiny and/or pressure from others to eat more, and a loss of control over diet. For example, one participant describes how their relationship is impacting upon their ED:

My relationship with my partner significantly impacts my eating behaviours and this has been worse since the pandemic.

Whilst hiding their ED from others left some participants reporting feelings of being "trapped", "incredibly low" and "unhappy":

I'm trapped at home with people who don't know I have anorexia. I am hiding and lying constantly.

When describing the challenges associated with being in lockdown with partners or family members, participants tended to refer to mealtimes. Specifically, participants felt that the people they lived with were often critical of their eating behaviours:

I have much less choice of what I eat and I have to eat most of my meals in front (and in scrutiny) of others and that is causing enormous stress.

This pressure to eat appears to be worsened by others worrying that the individual will be unable to recover if they contract COVID-19, due to not correctly fuelling their body:

My mother is particularly concerned about me because I "do not give my body enough energy to overcome the virus" if I contracted it, and she wants me to eat more because of this. This is weighing on my mind because I absolutely do not want to do this.

Suddenly living with others resulted in many participants losing control over the meals they were able to eat, for example due to others taking responsibility for cooking:

My partner is making more of our meals so I'm less certain about the ingredients and amounts, so less able to track this using MyFitnessPal. I'm also less in control of what I eat for breakfast and lunch, when usually I would have full control over this in the office.

In the likely event of the pandemic continuing for some time, the prolonged stress of living in challenging environments could be particularly detrimental for those experiencing EDs.

## Social isolation and reduced access to usual social support networks

The vast majority (86.4%) of our sample reported greater feelings of social isolation as a result of the pandemic (Table 3). For many individuals, spending time with friends and family represents a vital factor in their ED recovery. Lockdown has resulted in many participants being unable to access and/or physically engage with these support networks. Impacts of this include worsened ED symptoms and reduced motivation for recovery:

Unable to see friends, who are supportive and boost my mood (which affects my eating).

The social aspect of eating can be particularly important for helping some individuals manage their symptoms. Going 'out' for food can provide an important coping mechanism to encourage them to eat. The lockdown clearly impedes this:

The MAJORITY of my "safe" eating takes place outside of the house. E.g. getting coffees, going to Pret for food I am familiar and comfortable with. There is something less shameful and "holding" for me about eating out, and that isn't the case at home. Now I have to eat more at home, on my own, and the isolation around eating and exercise is not a good place for me to be.

Unfortunately, isolation can also impact recovery through reduced feelings of accountability. Whilst individuals who find themselves suddenly living with others may experience increased feelings of scrutiny and pressure to recover (refer to Disruption to living situation section), the opposite can occur for those who find themselves isolated from others. Over 20% of our sample reported feeling less social pressure to recover due to the pandemic:

Not seeing those close to me who would recognise my losing weight and deterioration has put me under less pressure to challenge my ED as I can unwitnessed lose weight without challenge from others which is less pressure for me.

Some participants felt that they were becoming more restrictive with their diet and expressed concerns around the impacts that this could have, not only now, but to their future recovery:

I am concerned that I am becoming more rigid about what I eat when I had moved on from this. I worry it will be harder to go back to social eating.

Here, we are provided with a crucial reminder that the impacts of the pandemic are likely to be felt for a prolonged period of time; continuing after cessation of the lockdown and the return to 'normality'. It is also important to recognise that ED behaviour that occurred during the pandemic may have been reinforced by repetition during a lengthy lockdown period. These reinforced or 'habitual' behaviours can be resistant to change.

#### Changes to physical activity rates

Participant responses to the impact of the pandemic upon physical activity were varied. The researchers initially assumed that physical activity rates may drop as a result of lockdown, and indeed almost half of our sample reported a reduction in physical activity. However, 36.5% reported an *increase* in physical activity (Table 3). Qualitative responses indicate that activity increases were primarily driven by anxieties about weight gain and a desire to counteract the effects of inactivity during lockdown:

I was doing lots before but even more now partly because I actually like it, partly to fill the time, partly from an unhealthy driven mentality. Rest days now feel unnecessary because I'm resting every day if I'm not working.

Not being able to freely engage in ED behaviours such as bingeing and purging provided another driver:

Now people are home with me my binge purge has reduced to every other day. However, my exercise has increased to compensate for this.

It can be easier for others to monitor or 'police' eating behaviours than it is to identify when an individual has crossed the line between healthy exercise and excessive, detrimental activity levels. It can also be easier for individuals to defend and rationalise physical activity, especially during lockdown when daily exercise was so widely discussed in government daily COVID-19 updates. Participants were aware of this tension between healthy activity and over-exercise, and reported finding it difficult to manage their relationship with exercise during lockdown:

There is part of me that wants to do all of the exercise, to 'get thin' and to listen to the disorder. But there is also part of me that knows I need to moderate my exercise because it's so triggering for me.

For those who did report a *decrease* in activity, the same anxieties around weight gain provided a driver for engaging in restrictive eating behaviours:

Can't exercise as much / the way I was, and not getting as many steps, so being more restrictive with foods eaten, calories, and timing of eating (doing intermittent fasting again).

This demonstrates that the pandemic can increase symptoms both in relation to eating behaviours and/or excessive exercise, depending upon the individual and the context they find themselves in during lockdown (Table 3).

#### Reduced access to healthcare services

One of the major challenges faced by participants was a reduction in healthcare service provision and/or discrepancies in access to healthcare services. Participants reported being prematurely discharged from inpatient units, having treatment suspended or remaining on a waiting list for treatment, and receiving limited post-diagnostic support:

The pandemic meant that I was discharged suddenly and prematurely from inpatient treatment, and has also meant that my post-discharge support is limited.

It was clearly a distressing time for participants, and a reduction in service provision caused some participants to feel like a "burden", an "inconvenience", and "forgotten" by the government and NHS. The consequences of not being able to access professional treatment during

Table 3 Reported changes due to COVID-19 pandemic and effect on ED symptoms

| Factor                   | Changed due to pandemic     | Ν   | %    | Effect of change on symptoms (of those who indicated change) | Ν   | %    |
|--------------------------|-----------------------------|-----|------|--|-----|------|
| Living situation         | Yes                         | 27  | 20.9 | Symptoms much worse  | 1   | 3.7  |
|                          | No                          | 98  | 76   | Moderately worse   | 11  | 40.7 |
|                          | Other                       | 4   | 3.1  | Slightly worse   | 11  | 40.7 |
|                          | Total                       | 129 | -    | No change  | 2   | 7.4  |
|                          |                             |     |      | Slightly better  | 1   | 3.7  |
|                          |                             |     |      | Moderately better  | 1   | 3.7  |
|                          |                             |     |      | Symptoms Much better   | 0   | 0    |
|                          |                             |     |      | Missing  | 0   | 0    |
|                          |                             |     |      | Total  | 27  | -    |
| Social isolation         | Much more isolated          | 56  | 43.4 | Symptoms much worse  | 33  | 28.2 |
|                          | Moderately more isolated    | 35  | 27.1 | Moderately worse   | 45  | 38.5 |
|                          | Slightly more isolated      | 17  | 13.2 | Slightly worse   | 24  | 20.5 |
|                          | No change                   | 8   | 6.2  | No change  | 12  | 10.3 |
|                          | Slightly less isolated      | 5   | 3.9  | Slightly better  | 2   | 1.7  |
|                          | Moderately less isolated    | 4   | 3.1  | Moderately better  | 0   | 0    |
|                          | Much less isolated          | 0   | 0    | Symptoms Much better   | 0   | 0    |
|                          | Missing                     | 4   | 3.1  | Missing  | 1   | 0.9  |
|                          | Total                       | 129 | -    | Total  | 117 | -    |
| Usual support network(s) | Yes                         | 93  | 72.1 | Symptoms much worse  | 14  | 15.1 |
|                          | No                          | 31  | 24   | Moderately worse   | 26  | 28   |
|                          | Missing                     | 5   | 3.9  | Slightly worse   | 27  | 29   |
|                          | Total                       | 129 | =    | No change  | 23  | 24.7 |
|                          |                             |     |      | Slightly better  | 3   | 3.2  |
|                          |                             |     |      | Moderately better  | 0   | 0    |
|                          |                             |     |      | Symptoms Much better   | 0   | 0    |
|                          |                             |     |      | Missing  | 0   | 0    |
|                          |                             |     |      | Total  | 93  | -    |
| Physical activity        | Much less physical activity | 36  | 27.9 | Symptoms much worse  | 41  | 36.9 |
|                          | Moderately less             | 15  | 11.6 | Moderately worse   | 32  | 28.8 |
|                          | Slightly less               | 13  | 10.1 | Slightly worse   | 28  | 25.2 |
|                          | No change                   | 13  | 10.1 | No change  | 8   | 7.2  |
|                          | Slightly more               | 16  | 12.4 | Slightly better  | 1   | 0.9  |
|                          | Moderately more             | 13  | 10.1 | Moderately better  | 0   | 0    |
|                          | Much more                   | 18  | 14   | Much better  | 1   | 0.9  |
|                          | Missing                     | 5   | 3.9  | Missing  | 0   | 0    |
|                          | Total                       | 129 | _    | Total  | 111 | _    |

the pandemic could be severe, and in some cases even fatal. These concerns were echoed by one participant:

People are going to die during this from their ED because they are unable to be physically monitored.

Unfortunately, this reduced physical monitoring links back to the aforementioned lack of accountability

(refer to Social isolation and reduced access to usual social support networks section) with participants reporting feeling greater 'opportunity' to engage in undetected ED behaviours:

Because I no longer can go to the day service, my anorexia makes me believe it's a chance to 'get away' with me e.g. losing a ton of weight.

Despite the reduction in service provision, some participants stated that they were still able to access professional healthcare services via technology, such as video-calling software. However, whilst online support was described as a positive factor, participants described this as falling short of treatment and support received in-person. They did not view online treatment as a direct replacement for face-to-face support, but rather the 'next best thing' or the 'only alternative' when traditional support mechanisms were unavailable:

I wouldn't usually access phone support or video chat. I dislike doing these, but felt I had to having lost my face to face support.

Unfortunately, participants reported discrepancies in access to services. Not all services are using technology to support individuals during the pandemic – or they have been delayed in their provision of such services. One participant described the inequality in service provision across the UK as a "postcode lottery":

There is a postcode lottery in terms of what services are now providing, some patients in some areas were offered video chat straight away, my service still haven't got this up and running.

Remote services also raised their own, perhaps unexpected, issues. For example, some participants felt that video calling was having a detrimental effect on their ED. Seeing their own video during the call, made some participants more aware and more critical of their appearance:

Have seen myself more and made me question if I good enough or look good. Also, things have changed to online video conferencing. It's just the amount of time seeing yourself.

It is vital that researchers and service providers accurately assess the appropriateness of technologies for remote service provision.

Some participants were also concerned about being asked to self-monitor by healthcare professionals, whilst relying on remote treatment. Many individuals experiencing EDs choose not to keep weighing scales in their homes as this can lead to an unhealthy fixation on their weight. This is in direct conflict with being asked to self-monitor to enable remote treatment:

Many of us trying to recover have gotten rid of our scales as for some they fuel the compulsive nature of the ED - I wonder how many will worsen due to being told to buy scales from ED services.

The challenges of physical monitoring - the ED service that I am under would prefer for me to have own scales and self-report weight whereas GP sees this as a big risk in itself (I would become dangerously fixated on the number).

Our findings highlight discrepancies in service provision, a lack of consistency in advice between services, concerns around the suitability of some remote treatment methods, and the need to develop clear evidence-based guidelines for best practice.

#### Disruption to routine and perceived control

For many, keeping to a regular routine is vital for ED recovery and relapse prevention. Unfortunately, many routines have been significantly impacted as a result of lockdown:

Feeling frustrated because I've been learning how to control the bulimic symptoms and normally able to manage them. Yet the change in routine (or lack of) and general stress/anxiety is unsettling it and noticed my thoughts and behaviours changing.

The majority of participants (65.9%) reported some change to their usual coping mechanisms as a result of the pandemic. A lack of routine and/or distractions created more time for rumination about weight, exercise habits, and meals. Routine can also contribute to an individual's perceived sense of control; another significant factor in recovery and general psychological wellbeing. When perceived control is low (such as during the pandemic), disordered eating can represent one thing that the individual *can* control:

Normally when I am in control over everything I am able to do better. I currently have less control of being able to go out, my studying, work etc and I feel subconsciously I am controlling this with food.

Due to the pandemic's overwhelming impact on routines, we may see an increase in ED behaviours as a coping mechanism.

#### Increased exposure to triggering messages

Unsurprisingly, given the current lockdown, the majority (81.4%) of participants reported spending more time online. This is a trend seen in the population more generally, with digital spaces switching from an amenity to a necessity . When asked whether this had an effect upon their ED symptoms, over half (55%) of our sample reported that this increase in time online had worsened their symptoms (Table 4).

**Table 4** Changes to time spent online due to the COVID-19 pandemic

| Time spent online    | N   | %    | Effect on ED symptoms <sup>a</sup> | N   | %    |
|----------------------|-----|------|------------------------------------|-----|------|
| Much more time       | 47  | 36.4 | Symptoms much worse                | 11  | 9.9  |
| Moderately more time | 34  | 26.4 | Moderately worse                   | 24  | 21.6 |
| Slightly more time   | 24  | 18.6 | Slightly worse                     | 27  | 24.3 |
| No change            | 13  | 10.1 | No change                          | 44  | 39.6 |
| Slightly less time   | 3   | 2.3  | Slightly better                    | 4   | 3.6  |
| Moderately less time | 3   | 2.3  | Moderately better                  | 0   | 0    |
| Much less time       | 0   | 0    | Much better                        | 0   | 0    |
| Missing              | 5   | 3.9  | Missing                            | 1   | 0.9  |
| Total                | 129 | -    | Total                              | 111 | _    |

<sup>a</sup>excludes those participants who reported 'no change' for time spent online

There are numerous reasons why time online could have a negative impact for those experiencing an ED. For example, it is possible that without their usual social support networks and professional healthcare services, individuals may find support from less helpful sources, encounter extreme content (such as pro-anorexia content) or come across misinformation online [7]. This was supported by some participants mentioning that they accessed pro-ana websites and/or searched for "thinspo" content. However, for many participants, the main contributing factors appeared to be the increase in food and exercise-related messages dominating public discourse. For example, many social media posts refer to the general public's "fear of gaining weight" during lockdown. These messages can be particularly triggering for individuals living with EDs:

There is a lot going around social media such as putting on weight during COVID-19 due to not doing as much and that is really triggering for people like myself as straight away my head thinks, that can't happen to me! I must change the way I am towards food etc.

There has also been an increase in posts about physical activity, for example, people sharing information on their 'daily exercise' or sharing content to encourage others to stay active at home (including increased availability of workout content, e.g., the free daily YouTube workouts delivered by 'body coach' Joe Wicks). Excessive exercise is a major factor in many EDs and participants reported finding it difficult to cope with this increased emphasis around physical activity:

There is a heavier push on ALL social platforms to 'stay for' and 'stay in shape'. Any coping mechanisms I did have before have changed. I struggle with exercise mainly, and with everything being in the spotlight it's harder to look away from it and ignore it.

Content promoting exercise, also caused participants to engage in social comparisons (e.g., comparing their body to others, or comparing their level of exercise):

Watching work out videos judging myself for my lack of motivation to do anything and just how disgusting my body is compared to everyone else's out there. Everyone is doing self-improvement and I'm not.

These messages not only dominated social media, but also mass media stories:

I think the messages we are given through the media right now are that by eating "healthy" and "keeping active" we are somehow protecting ourselves from the virus. Actually, I don't think there is sufficient, scientifically valid evidence that this is the case. I can be rational about this, but nonetheless, I find these constant reminders quite triggering.

Whilst public health initiatives that promote healthy eating and exercise might be beneficial for some members of the general public, it is clear from participants' statements that a focus on exercise and food can be triggering for people with EDs. This is particularly important when screen time is likely to be increased due to lockdown.

#### Changes to relationship with food

The lockdown has resulted in a change in food behaviours across the population, for example we have seen evidence of food hoarding leading to supply shortages. Of the participants in our study, 66.7% reported that their relationship with food has changed due to the pandemic. Lockdown can result in individuals having a shortage of food in the house (or provide them with a reason to justify a lack of supplies), or excess amounts if they are in a household that has stocked up. Any changes to food behaviours have the potential to have a significant impact upon individuals with experience of EDs. We can see some of the range of impacts in our data. Some participants raised concerns about being more likely to binge during the pandemic due to increased access to food in the household:

Being at home makes me more likely to binge as there is more food in the house.

Participants also reported increased levels of rumination over their food intake, which they attributed to its "scarcity" and being unable to purchase "safe foods", referring to food that is often low in calories that a person feels is safe to eat:

I feel like I am thinking more and more about food. Part of it is scarcity and the inability to eat healthy food - this has made me feel a bit trapped and like it's all I can focus on.

A perceived scarcity of food during the pandemic caused some participants to feel guilty for eating food. These participants said they felt "less deserving" of food than other people:

Restriction can become a 'reasonable' thing to do because you're saving food for later or someone else.

For these participants, the pandemic reinforced their ED behaviours by providing them with a way of rationalising food restriction. In addition to experiencing increased feelings of guilt, many participants also expressed that their relationship with food had become more negative and purely functional as a result of the pandemic:

Totally unenjoyable; purely functional, seeing as a 'prescription medication', a means to an end, almost resentful of food.

Interestingly, some participants used the pandemic to rationalise their food restrictions for other reasons stating that they felt they were "less likely to get sick with anything if I don't eat". This is a stark contrast from the attitudes they reported from friends and family who encourage them to eat to prevent getting sick (see Disruption to living situation section).

#### Positive outcomes

It was uplifting to also see positive signs of adaptive coping mechanisms within our data, primarily in response to the use of technology to access support. Due to reduced physical access to usual support networks, some participants reported using the internet and social media (e.g., Facebook groups, WhatsApp, and YouTube) to speak to friends, access support from ED communities, and to follow others' journeys to/of recovery. These platforms can represent an important resource during periods of isolation. Social media appears to facilitate social connection and support, thus combatting some of the challenges of the pandemic:

Looking at support networks for bulimia and talking to friends who know I'm fearing a relapse due to this pandemic. Some participants identified unexpected, positive impacts of the pandemic. For example, for some individuals the restrictions on ED behaviour attributed with changes in their living situation were identified as a positive:

Binging [sic] and purging has reduced. I was doing this 5 times a day. Now people are home with me my binge purge has reduced to every other day.

This stands in stark contrast with many participants' experiences, who found changes in living situation to be incredibly stressful (see Disruption to living situation section).

Others found positives in the social distancing imposed by lockdown. For example, explaining that reduced social interaction also reduced social comparisons, thereby reducing anxiety:

Fewer social interactions with others maybe mean fewer self-comparisons with others & also not experiencing same level of anxieties about being 'at fault' in some way

Others reflected on their choice of clothing during the lockdown period, reporting that they wore more comfortable clothing that was less triggering of their symptoms:

I have been less critical of and aware of my body and its day to day changes. This is because I have been wearing only leggings or tracksuits and yoga bras and haven't been 'triggered' by uncomfortable clothing -previously my Jeans would dig in after lunch and I would fall into a spiral of thinking I'd gained weight/body checking etc.

However, it should be noted that these individuals are not representative of the vast majority of participants, with many experiencing increased pressures and stress as a result of the pandemic. Also some positives, such as those associated with a lack of social interaction, may not represent a healthy long-term solution.

#### **Discussion**

Our findings highlight eight key themes describing how the COVID-19 pandemic is affecting individuals with experience of EDs: (1) Disruption to living situation; (2) Increased social isolation and reduced access to usual support networks; (3) Changes to physical activity rates; (4) Reduced access to healthcare services; (5) Disruption to routine and perceived control; (6) Changes to relationship with food; (7) Increased exposure to triggering messages; and (8) Positive outcomes.

In keeping with previous research, our findings suggest that patients (and their loved ones) may be grateful of opportunities for remote treatment and support when inperson alternatives are not available [5, 10] – something also suggested in the recent COVID-19 call to action [3]. Preliminary findings suggest that health treatment delivered remotely via technology may be effective for those experiencing EDs, however it is noted that there is still limited evidence in this regard and more research is necessary [5]. It is also noted that serious consideration must be given beyond digital interventions due to potential inequalities in access to technology [3].

Alarmingly, our data shows inequalities in UK health service provision during the pandemic, this must be addressed to ensure consistency in service provision across the UK. In addition to ensuring equality of access, our findings highlight other barriers to effective remote treatment, such as how to monitor health and recovery progress without having a detrimental impact on mental wellbeing. This includes not requiring weighing scales within the individual's household, and providing alternatives to technologies, such as video-calling software, which may inadvertently make users critical of their appearance (this supports previous findings (e.g., [8, 10]). In other instances, software may have the required functions available - such as an option to 'hide self-view' - but users may not be aware of this; increasing awareness or setting this option as a default should be considered. Clear guidelines are required for the design and implementation of appropriate remote/digital health services.

Our findings demonstrate the importance of social support from friends and family. Social relationships can have a protective buffering effect from stressful events [3]. This protective effect can be diminished during lockdown periods. Future research should look to identify ways to minimise the detrimental effects of physical distancing, and develope and/or encourage alternative positive coping mechanisms.

Media coverage and social media posts were a source of anxiety for our participants, due to a public preoccupation with food, weight gain and exercise. Although positive messages about diet and exercise can be beneficial for the majority of the population, it is important for healthcare and government to acknowledge that these can be triggering or upsetting for vulnerable populations, and work to establish guidelines to help address this. Risk-elevating messages designed to nudge individuals towards social distancing and safe behaviour can also lead to increased anxiety and reduced feelings of control over the situation; again, factors linked to negative impacts on psychological wellbeing and ED symptoms. As described by Holmes et al. [3] 'research is needed to inform future approaches, including strategies to help individuals to stay informed by authoritative sources, prevent over-exposure to media, and mitigate and help manage the effect of viewing [ ...] traumatic content' (p. 552). Our findings suggest a need for clear guidelines driven by the insight of those with lived experience - around media reporting, social media policies and positive strategies to manage emotional and psychological consequences.

Crucially, our findings illustrate why we must not underestimate the longevity of the impact of the pandemic. Individuals with experience of EDs will likely experience a long-term effect on their symptoms and/or recovery. It is important that this is recognised by healthcare services, and beyond, in order to offer the necessary resources to support this vulnerable population now and on an on-going basis.

The lessons learned as a consequence of the COVID-19 pandemic have the potential to be relevant to other public health emergencies, other ED populations, and to future circumstances which involve periods of lockdown, food shortages, and/or social isolation. As there are international studies in progress (e.g., [11]), crosscultural comparisons could be an interesting area for future research – particularly in relation to how different countries have addressed ED treatment and healthcare during, and after, the pandemic.

#### Limitations

While this study provides an in-depth examination of the impact of the pandemic on the lives of people with EDs, it does have some limitations. Firstly, all participants were recruited via social media, resulting in the sample being biased toward those who have access to and are familiar with technology. This opportunistic approach may also have resulted in the sample being biased towards people who are currently experiencing difficulties, rather than those who are not facing challenges related to their EDs. Secondly, information about specific diagnoses was not collected. This information could be collected in future research to enable comparisons between different groups of people experiencing EDs.

Finally, data were only collected at a single time point during the early stages of the UK lockdown. The pandemic will likely continue for some time, potentially having long-lasting impacts upon the lives of individuals experiencing EDs. Thus, longitudinal research is required to investigate the long-term effects of the pandemic on this population.

Due to not having a baseline measure (i.e., prepandemic) it is not possible to make assumptions about the impact of the pandemic on the quantitative scale scores within our sample. Again, longitudinal research could be beneficial in this regard - allowing the comparison of data from the start of lockdown with data from later time points. This would help identify the ongoing, and potentially evolving, impact of the pandemic as it progresses. The researchers are planning a follow-up study to address this. Future research may also wish to include further quantitative analysis of the relationships

between the facilitators (e.g., low perceived control) and barriers to ED (high social support) behaviour during periods of lockdown.

#### **Conclusions**

In summary, this research suggests that the ED population is at significant risk of negative impacts of the pandemic; the consequences of which may be felt long after a societal return to 'normality'. There is a vital need for interventions to support this population during the current pandemic and beyond. Interventions should be co-designed with end-users to facilitate effective treatment. Our findings also highlight inequalities in health-care provision that must be addressed; a more cohesive approach to remote treatment is required across all UK healthcare services. Positive aspects of technology use were identified but it is also necessary to consider potential negative impacts of public messages around food and exercise behaviours.

#### **Supplementary information**

**Supplementary information** accompanies this paper at https://doi.org/10. 1186/s40337-020-00319-y.

**Additional file 1: Supplementary Material 1.** Copy of survey items. Copy of all items included in the online survey for this study

#### Abbreviations

COVID-19: Coronavirus Disease 2019; ED: Eating Disorder; ESSI: ENRICHD Social Support Instrument; NHS: National Health Service; PSS: Perceived Stress Scale; RRS-ED: Rumination Response Scale for Eating Disorders; SCI: Shapiro Control Inventory; WEMWBS: Warwick-Edinburgh Mental Wellbeing Scale

#### Acknowledgements

Not applicable.

#### Authors' contributions

DB led initial idea formation and project management. Both authors were involved in designing the survey and data collection. DB led on quantitative analysis; CT led on thematic coding of qualitative data. Both authors played a major role in the write up of the results. DB led on initial preparation of the manuscript. The author(s) read and approved the final manuscript.

#### Funding

No funding was obtained for this research.

#### Availability of data and materials

The anonymised datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.

#### Ethics approval and consent to participate

Ethical approval for this research was granted by the Northumbria University ethics board [23259]. All participants gave informed consent.

#### Consent for publication

Not applicable.

#### Competing interests

The authors declare that they have no competing interests.

Received: 21 May 2020 Accepted: 9 August 2020 Published online: 24 August 2020

#### References

- Rodgers RF, Lombardo C, Cerolini S, Franko DL, Omori M, Fuller-Tyszkiewicz M, et al. The impact of the COVID-19 pandemic on eating disorder risk and symptoms. Int J Eat Disord. 2020;53:1166–70. https://doi.org/10.1002/eat.23318.
- Yao H, Chen JH, Xu YF. Patients with mental health disorders in the COVID-19 epidemic. Lancet Psychiatry. 2020;7:e21. https://doi.org/10.1016/S2215-0366(20)30090-0.
- Holmes EA, O'Connor RC, Perry VH, Tracey I, Wessely S, Arseneault L, et al. Multidisciplinary research priorities for the COVID-19 pandemic: a call for action for mental health science. Lancet Psychiatry. 2020;7:547–60. https:// doi.org/10.1016/S2215-0366(20)30168-1.
- Davis C, Chong NK, Oh JY, Baeg A, Rajasegaran K, Elaine Chew CS. Caring for children and adolescents with eating disorders in the current COVID-19 pandemic: a Singapore perspective. J Adolesc Health. 2020. https://doi.org/ 10.1016/j.jadohealth.2020.03.037.
- Gordon CM, Katzman DK. Lessons learned in caring for adolescents with eating disorders: the Singapore experience. J Adolesc Health. 2020. https:// doi.org/10.1016/j.jadohealth.2020.03.041.
- Manasse SM, Schumacher LM, Goldstein SP, Martin GJ, Crosby RD, Juarascio AS, et al. Are individuals with loss-of-control eating more prone to dietary lapse in behavioural weight loss treatment? An ecological momentary assessment study. Eur Eat Disord Rev. 2018;26:259–64. https://doi.org/10. 1002/erv.2583.
- Branley DB, Covey J. Pro-ana versus pro-recovery: a content analytic comparison of social media users' communication about eating disorders on twitter and Tumblr. Front Psychol. 2017;8:1356. https://doi.org/10.3389/ fpsyq.2017.01356.
- Murphy R, Calugi S, Cooper Z, Dalle GR. Challenges and opportunities for enhanced cognitive behaviour therapy (CBT-E) in light of COVID-19. Cogn Behav Ther. 2020;13. https://doi.org/10.1017/S1754470X20000161.
- Waller G, Pugh M, Mulkens S, Moore E, Mountford VA, Carter J, et al. Cognitive-behavioral therapy in the time of coronavirus: clinician tips for working with eating disorders via telehealth when face-to-face meetings are not possible. Int J Eat Disord. 2020;53:1132–41. https://doi.org/10.1002/ eat.23289.
- Fernández-Aranda F, Casas M, Claes L, Bryan DC, Favaro A, Granero R, et al. COVID-19 and implications for eating disorders. Eur Eat Disord Rev. 2020;28: 239–45. https://doi.org/10.1002/erv.2738.
- Phillipou A, Meyer D, Neill E, Tan EJ, Toh WL, Van Rheenen TE, et al. Eating and exercise behaviors in eating disorders and the general population during the COVID-19 pandemic in Australia: initial results from the COLLATE project. Int J Eat Disord. 2020;53:1158–65. https://doi.org/10.1002/eat.23317.
- Wong JEL, Leo YS, Tan CC. COVID-19 in Singapore current experience: critical global issues that require attention and action. J Am Med Assoc. 2020;323:1243–4. https://doi.org/10.1001/jama.2020.2467.
- Temorshuizen JD, Watson HJ, Thornton LM, Borg S, Flatt RE, MacDermod CM, et al. Early Impact of COVID-19 on Individuals with Eating Disorders: A survey of ~1000 Individuals in the United States and the Netherlands. MedRxiv Prepr Serv Heal Sci. 2020;2020(05):28.20116301. https://doi.org/10. 1101/2020.05.28.20116301.
- Tennant R, Hiller L, Fishwick R, Platt S, Joseph S, Weich S, et al. The Warwick-Edinburgh mental well-being scale (WEMWBS): development and UK validation. Health Qual Life Outcomes. 2007;5:1–13. https://doi.org/10.1186/ 1477-7525-5-63.
- NHS Health Scotland. Warwick-Edinburgh Mental Well-being Scale (WEMWBS) user guide. 2016.
- Cohen S, Kamarck T, Mermelstein R. A global measure of perceived stress. J Health Soc Behav. 1983;24:386–96. https://doi.org/10.2307/2136404.
- 17. Karam F, Bérard A, Sheehy O, Huneau M-C, Briggs G, Chambers C, et al. Reliability and validity of the 4-item perceived stress scale among pregnant women: results from the OTIS antidepressants study. Res Nurs Health. 2012; 35:363–75. https://doi.org/10.1002/nur.21482.
- Vaglio J, Conard M, Poston WS, O'Keefe J, Haddock CK, House J, et al. Testing the performance of the ENRI CHD social support instrument in cardiac patients. Health Qual Life Outcomes. 2004. https://doi.org/10.1186/ 1477-7525-2-2.

- Blumenthal J, Babyak M. Al. E. Enhancing recovery in coronary heart disease patients (ENRI CHD): study design and methods. Am Heart J. 2000;139:1–9. https://doi.org/10.1016/S0002-8703(00)90301-6.
- Shapiro Jr. DH. Shapiro control inventory (SCI) manual 1994. http:// controlresearch.net/shapiro-control-inventory-manual.html Accessed 4 May 2020.
- Cowdrey FA, Park RJ. Assessing rumination in eating disorders: principal component analysis of a minimally modified ruminative response scale. Eat Behav. 2011;12:321–4. https://doi.org/10.1016/j.eatbeh.2011.08.001.
- Smith KE, Mason TB, Lavender JM. Rumination and eating disorder psychopathology: a meta-analysis. Clin Psychol Rev. 2018;61:9–23. https://doi.org/10.1016/j.cpr.2018.03.004.
- Gardner B, Rebar AL, Gardner B, Rebar AL. Habit formation and behavior change. Oxford Res Encycl Psychol. 2019. https://doi.org/10.1093/acrefore/ 9780190236557.013.129.
- Beaunoyer E, Dupéré S, Guitton MJ. COVID-19 and digital inequalities: reciprocal impacts and mitigation strategies. Comput Human Behav. 2020; 106424. https://doi.org/10.1016/j.chb.2020.106424.
- Southey F. Panic buying amid coronavirus fears. [Online] Available at: https://www.foodnavigator.com/Article/2020/03/27/Panicbuyingamidcoronavirus-fears-How-much-are-we-spending-and-why-is-it-aproblem. Accessed 22 Apr 2020.
- Anderson KE, Byrne CE, Crosby RD, Le Grange D. Utilizing Telehealth to deliver family-based treatment for adolescent anorexia nervosa. Int J Eat Disord. 2017;50:1235–8. https://doi.org/10.1002/eat.22759.

#### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

#### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

